Author's response to reviews

Title: Ovarian carcinoma associated with pregnancy: a clinicopathologic analysis of 23 cases and review of the literature

Authors:

Nadereh Behtash (nadbehtash@yahoo.com)
Mojgan Karimi Zarchi (drkarimi2001@yahoo.com)
Mitra Modares Gilani (M_gilani@yahoo.com)
Fatemeh Ghaemmaghami (ftghaemmagh@yahoo.com)
Azamsadat Mousavi (Azamsadat_mousavi@yahoo.com)
Fahimeh Ghotbizadeh (f_ghotbi@yahoo.com)

Version: 5 Date: 9 January 2008

Author's response to reviews: see over
Hi. Many thanks for your message. Excuse me for the these missing points. I correct the reference number that was written in discussion. Also I changed the afflictions of the authors name. I must say you that my first name is Mojgan and my family name is Karimi Zarchi. I hope to accept these changes and help me for publication of this article in your well journal.